Author's response to reviews

Title: The wider social environment and changes in self-reported quality of life in the transition from late childhood to early adolescence: a cohort study

Authors:

Marjan Drukker (Marjan.Drukker@sp.unimaas.nl)
Charles Kaplan (charliekaplan@mac.com)
Josien Schneiders (J.Schneiders@sp.unimaas.nl)
Frans JM Feron (F_Feron@zzl-ggd.nl)
Jim Van Os (J.vanos@sp.unimaas.nl)

Version: 6 Date: 22 March 2006

Author's response to reviews:

We incorporated the requested changes